Author's response to reviews

Title: Fibulin-1 is Epigenetically Down-regulated and Associated with Bladder Cancer Recurrence

Authors:

Wei Xiao (xiaowei0041@163.com)
Ji Wang (snokey@gmail.com)
Heng Li (lihengtjmu@163.com)
Ding Xia (mickxia@126.com)
Gan Yu (cdyi1987@126.com)
Weimin Yao (Yaoweiimin1985@126.com)
Haibing Xiao (445829439@qq.com)
Bin Lang (langbindoctor@yahoo.com.cn)
Xin Ma (maxin20000mail@163.com)
Xiaolin Guo (guoxiaolin0488@sina.com)
Wei Guan (deniskwan@gmail.com)
Hua Xu (xuhua@mail.hust.edu.cn)
Jihong Liu (jhiu@tjh.tjmu.edu.cn)
Xu Zhang (xzhang@foxmail.com)
Zhangqun Ye (zhangqun_ye@yahoo.com.cn)

Version: 4 Date: 1 September 2014

Author's response to reviews: see over
Dear editor:
Thank you for your comments. We have carefully read these comments and revised manuscript according to them. Here is a point-to-point reply.

1. Please include email addresses for all authors on your title page.
   
   Reply:
   A section including email addresses of all authors has been added in the title page as requested.

2. Please format your abstract according to our guidelines: (http://www.biomedcentral.com/bmccancer/authors/instructions/researcharticle#formatting-abstract).

   Reply:
   The abstract has been revised into a structured form according to the guidelines.

3. You state that your animal study followed guidelines, however, please can you clarify whether your animal studies were approved by an animal ethics committee and provide the full name of the committee in your methods.

   -Please include a clearly titled conclusion section.
   -Please include a 'Competing interests' section between the Conclusions and Authors' contributions.

   Reply:
   A clarification has been included in the Methods-> Animal experiments.
   -A clearly titled “Conclusions” section has been included after the “Discussion”.
   -A “Competing interests” section has been moved from the title page to the middle of the two sections.

4. Please include an Authors' contributions section before the Acknowledgements and Reference list.

   Reply:
   An “Authors’ contributions” section has been included before the Acknowledgements and Reference list.

Thank you very much.

Sincerely yours,

Dr. Hua Xu